\begin{table}%t1
%\centering
\par
\caption{\label{table.V}$V$ magnitudes of V1247~Ori published in the literature.}
\begin{tabular}{cl c cl}
\hline \hline \noalign{\smallskip}
$V$ & Ref.\tablefootmark{a} & & $V$  & Ref.\tablefootmark{a} 	\\
{[mag]} &  &	& {[mag]}  		& \\
            \hline
9.88 $\pm$ 0.01:	& Sh62		&	& 9.82 $\pm$ 0.03	& TYC   \\ 
9.854 $\pm$ 0.007	& Ru76		&	& 9.87 $\pm$ 0.02	& Ri00  \\
9.846 $\pm$ 0.013	& Ru88		&	& 9.81 $\pm$ 0.05	& Fu02  \\
9.844 $\pm$ 0.010	& LR92		&	& 9.88 $\pm$ 0.01:	& Vi03	\\
9.90 $\pm$ 0.01		& Eg92		&	& 9.85 $\pm$ 0.13	& Dr06	\\
9.92 $\pm$ 0.01: 	& Ha99		\\  
\hline
\end{tabular}
\tablebib{\tablefoottext{a}{Sh62: Sharpless (\cite{Shar62});
Ru76: Rufener (\cite{Ruf76}); Ru88: Rufener (\cite{Ruf88}); LR92: Lampens \& Rufener (1992); Eg92: Egret et~al. (\cite{Egret92}); Ha99: Handler (\cite{Hand99}); TYC: H{\o}g et~al. (\cite{Hog00}), transformed from $B_T V_T$ using the expression $V = V_T - 0.090 (B_T - V_T)$; Ri00: Richmond et~al. (\cite{Rich00}); Fu02: Fujii et~al. (2002); Vi03: Vieira et~al. (\cite{Vieira03}); Dr06: Droege et~al. (2006).}}
%\vspace*{-2mm}
\end{table}